Author's response to reviews

Title: Age and manifestation related symptoms in familial adenomatous polyposis

Authors:

Roland S. Croner (Roland.Croner@chir.imed.uni-erlangen.de)
Wolfgang M. Bruckl (wbrueckl@molmed.uni-erlangen.de)
Bertram Reingruber (Bertram.Reingruber@chir.imed.uni-erlangen.de)
Werner Hohenberger (Werner.Hohenberger@chir.imed.uni-erlangen.de)
Klaus Guenther (Klaus.Guenther@chir.imed.uni-erlangen.de)

Version: 8 Date: 21 February 2005

Author’s response to reviews:

To whom it may concern,

the requested changes regarding the form of the manuscript were done:

1. The title of the manuscript is identical with the uploaded title
2. MD was deleted after the authors names
3. The complete addresses of all authors were added
4. The keywords were deleted in the manuscript front page
5. The tables were inserted in the manuscript
6. The names of all authors from the section references were included
7. Competing interest was changed in COMPETING INTERESTS
8. A section of AUTHORS’ CONTRIBUTIONS was inserted
9. The heading in the figures were deleted

Unfortunately we did not understand which changes were necessary for the section heading.

All headings were checked for an equal style. There are no sections which could be deleted.

Please don't hesitate to contact me for any further questions.

Sincerely

RS. Crone